Author's response to reviews

Title: Diagnosis of tuberculosis in an Indian population by an indirect ELISA protocol based on detection of Antigen 85 complex: a prospective cohort study

Authors:

   Rajpal S. Kashyap (raj_ciims@rediffmail.com)
   Anju N Rajan (anju_mudaliar@rediffmail.com)
   Sonali S Ramteke (sonaliramteke8@yahoo.co.in)
   Vijay S Agrawal (ciims_ngp@sancharnet.in)
   Sanjivani S Kelkar (ciims_ngp@sancharnet.in)
   Hemant J Purohit (hemantdrd@hotmail.com)
   Girdhar M Taori (taorigm_ciims@yahoo.co.in)
   Hatim F Daginawala (hfd_ciims@rediffmail.com)

Version: Date: 28 May 2007

Author's response to reviews:

Covering letter

To,

Dr Chrissie Kouremenou
Senior Assistant Editor
BMC Infectious Diseases,

Dear Dr. Chrissie

Please find enclosed, revised copy of the manuscript entitled "Diagnosis of tuberculosis in an Indian population by an indirect ELISA protocol based on detection of Antigen 85 complexes: a prospective cohort study". The Manuscript has been modified as suggested by referee. The quality of our paper has remarkable improved by the comments and suggestions.

With Best regards

Dr.H.F.Daginawala (PhD)
Sr. Research Consultant.
Central India Institute Of Medical Sciences
Nagpur-10.
India
Dated: 28/05/07